OODS Tutorial 1

© 1999/2000 Michael Golm Universität Erlangen Nürnberg, Institut für Informatik 4

Guidelines for the OODS Assignments

• Every student has a project home /proj/i4oods/<loginname>. • In your project home you have to create a subdirectory for each assignment, e.g. for the fifth assignment you have to execute:

mkdir /proj/i4oods/<loginname>/assignment5

• The solutionshave tobe submittedusingthe handinprogram,e.g. tohand inthe fifthassignment you have to execute:

/proj/i4oods/pub/handin assignment5

• You don’t have to hand in the assignments that are labeled with "0 points". • The solutions that you submit must be your own. Cheating will be punished. • Youmust followthe Java coding guidelines (you definitelyloose credit points ifyour code does not conform to these rules):

• Class names begin with an upper case letter.

• Method names begin with alower case letter.

• Variable names begin with alower case letter.

• Constant (final static) variables areall upper case.

• Variable names are expressive, indicating their purpose.

• If a specific program structure is described in the assignment, your solution must followthis description. Your program must use

• names of classes and packages

• visibility (protection) of variables and methods

• names, parameter types, and return values of methods

as described in the assignment.

• You will loose credit points, if

• you do not implement a required functionality

• you implement more functionality than required

• your program can not be compiled

• Yourcode must be clear and understandable. Ifit contains complicated parts, you should explain them with comments. Where appropriate you should use javadoc comments.

• If you encounter problems with the assignments, don’t hesitate to contact us at {golm,rastofer}@informatik.uni-erlangen.de

OODS Tutorial 2

© 1999/2000 Michael Golm Universität Erlangen Nürnberg, Institut für Informatik 4

Homework Assignment #1: Drawing Editor 0 points

The purpose ofthisassignment is to introduceyouto the Java programminglanguage and development environment.Youhave towritea simpledrawing-editorapplet that can displaycirclesand allows the user to move and delete these circles. From subtask a) through d) we will guide you through the process of writing this applet.

a) Write theDrawingEditor.html file, theDrawingEditor class, and the Circle class.

ACircle objectis a circlethat is characterized by its upper leftcorner and its diameter. The Circleshouldhave a methoddraw(Graphics g) that ispassed a Graphicsobject.Thedraw method uses this Graphics object to draw itself. An instance of DrawingEditor should hold up to 100 Circle objects in an array. In its paint() method theDrawingEditor should draw all circles by calling their draw method. Totest the applet, you initially(inthe init() method) should create some circlesand insert them into the array.

b) Nowit’stimeforsome user interaction.We interactivelywant to move the shapes. Hence the

DrawingEditor has to react to mouse movements. Implementthe mouseDrag(Event
e, int x, int y) method in your applet class. (The compilerissues warnings about

"deprecated"methods. Please ignorethese warnings.)Findthe shape, the point(x,y)is located in,and set the positionofthisshape to (x,y).Forthispurpose, the Circle should provide a move(int x, int y) method. Finally you have to initiate a repaint of the applet.

c) Movingthe circlebysimplysettingits(x,y)positionmakes the circlejumptothe mouse pointer positionwhen itstarts to move. Abetter solutionis to findthe mouse pointeroffsetwithinthe shape when itis firsttouched. DrawingEditorhas to react to the mouseDownevent and compute the offsetwithinthisevent-handlermethod. The mouseDrag()method then subtracts the offsetfromthe (x,y)coordinates. Use instance variables ofDrawingEditor to pass information between the different event-handler methods.

d) Auser should be able to delete circlesbyholdingdownthe Shiftkeywhileclickingon a circle.

Ifthe mouse is withinthe area ofmore than one circle,the drawingeditordeletes allofthem.